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o Theory
o Seismometer analysis(Francesca)
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KAGRA schedule
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- We started the refurbishment work from this month




RSE tasks

- DRMI locked more the 30min with 3f signals

- Remaining tasks were CARM, DARM control with IR and transient to 1f

- Because of the schedule of the blackout(14th Oct.) and opening chamber
(next week), we postponed the interferometer commissioning to next year
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Blackout 14th Oct.

- The detalil will be explain by Washimi-san
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IYC experiment

- Purpose
- Establish the cooling procedure
- Performance check of nhew moving mass system(alignment)
- Vibration analysis of IYC
- Schedule
- Oct. - Nov. Preparation
- Dec. Cooling down
- Jan. Experiment

- Help the characterization using PEMs
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- PTEPO?Z : Noise budget O3GK : Analysis is ongoing
- PTEPOS3 : Overview of CAL, DET, PEM, GIF submitted Prellmf i
- arXiv2009.09305 L "A
- Acoustic injection paper : 1st draft e e e w
- Explained by Washimi-san later
- ICA for E-run/O3GK : Analysis is ongoing
- Portable PEMs : Writing
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Table 2 Summary of the KAGRA PEM sensors installed for the O3CK ol
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PEM updates

- Set the magnetometer at the outside of the mine
- We will discuss in this meeting
- Set trillium compact(small seismometer) somewhere
- Welcome the idea of position
- Preparing for the portable PEMs
- We bought the BNC power driver
- Preparing the injection items for both end station
- They were In only center area.



Next step

- Clean the PEMs cabling and setting
- Remove the BNC type cabling
- Shorten the cable length
- New PEM configuration from O3GK
- (Reminder) KAGRA PEM map for O3GK

- Clean up KAGRA ground

- Check the water fluid
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